Author’s response to reviews

Title: Adherence to antihypertensive medication in Russia: a scoping review of studies on levels, determinants and intervention strategies published between 2000 and 2017

Authors:
Elena Bochkareva (ebochkareva@gnicpm.ru)
Ekaterina Butina (ebutina@gnicpm.ru)
Irina Kim (ivkimivkim@gmail.com; ikim@gnicpm.ru)
Anna Kontsevaya (koncanna@yandex.ru)
Oksana Drapkina (odrapkina@gnicpm.ru)
David Leon (david.leon@lshtm.ac.uk)
Martin McKee (martin.mckee@lshtm.ac.uk)

Version: 2 Date: 16 Aug 2019

Author’s response to reviews:

Response to the comments of the editor:

The manuscript can be accepted after some minor editorial changes:
$ include in the title the period covered by the scoping review

We agree with the reviewer and we have made this change.

$ include in the title of all tables (except additional table 1) and the graph information on the subject, country and period.

We agree with the reviewer and we have made these changes.